Author's response to reviews

Title: Exploring the interpersonal-, organization-, and system-level factors that influence the implementation and use of an innovation -- synoptic reporting -- in cancer care

Authors:

Robin Urquhart (urquhartr@cdha.nshealth.ca)
Geoffrey A Porter (geoff.porter@dal.ca)
Eva Grunfeld (eva.grunfeld@utoronto.ca)
Joan Sargeant (joan.sargeant@dal.ca)

Version: 2 Date: 30 December 2011

Author's response to reviews:

Dear Dr. Wilson,

Please find the revised Study Protocol uploaded, with a revised Background section that makes a stronger case for the evidence base behind synoptic reporting.

Thank you once again for the opportunity to resubmit this manuscript. If you have any questions regarding the revised manuscript, do not hesitate to contact me.

Warm Regards,
Robin Urquhart